Author's response to reviews

Title: Progesterone after previous preterm birth for prevention of neonatal respiratory distress syndrome (PROGRESS): a randomised controlled trial.

Authors:

Jodie M Dodd (jodie.dodd@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)
Andrew J McPhee (mcpheea@wch.sa.gov.au)
Vicki Flenady (vicki.flenady@mater.org.au)
Jeffrey S Robinson (jeffrey.robinson@adelaide.edu.au)

Version: 3 Date: 19 February 2009

Author's response to reviews:

As requested we have addressed the following changes to the manuscript.
1) The abstract has been structured into background, methods, and discussion
2) The affiliations of each author have been formatted correctly
3) We have added a short discussion section after the methods to indicate that this is a trial protocol
4) The full details for reference 5 have been provided
5) Title has been removed from the title page
6) There are no author qualifications on the title page
7) Underlining has been removed from the manuscript
8) Headings have been changed as requested